

The quality of life of frequently vs. infrequently screened HPV vaccinated women

Kaisa Taavela^{1,2} · Tiina Eriksson³ · Heini Huhtala⁴ · Anne Bly³ · Katja Harjula³ · Kaisa Heikkilä³ · Mari Hokkanen³ · Mervi Nummela³ · Laura Kotaniemi-Talonen^{1,2} · Matti Lehtinen⁵ · Karolina Louvanto^{1,2}

Accepted: 27 November 2023 / Published online: 18 January 2024 © The Author(s) 2024

Abstract

Purpose Cervical lesions caused by human papillomavirus (HPV) are related to decreased quality of life (QoL) of women. Also, cervical cancer (CC) screening can cause psychological adverse effects. It has been assumed that by decreasing the HPV-related disease burden, HPV vaccinations would increase the QoL. This study compares the effect of CC screening on QoL of HPV vaccinated women in two different screening protocols.

Methods A total of 753 HPV16/18 vaccinated women were randomized to frequent (22/25/28 years of age) and infrequent (28 years of age) CC screening arms. QoL questionnaires (EQ VAS, RAND 36, amended CECA 10) were sent at the age of 28.

Results Median EQ VAS scores were 80 (Q_1 – Q_3 75–90) in both screening arms. Mean RAND 36 scores of frequently and infrequently screened women were 78.13/81.64 in Physical role functioning domain and, respectively, 77.93/80.18 in Pain, 69.10/69.12 in General Health, 54.67/53.61 in Energy, 83.72/85.11 in Social functioning, 69.53/69.68 in Emotional role functioning, and 68.16/69.29 in Emotional well-being domain. Among women with a self-reported history of Pap cytology abnormalities, overall mean scores of amended CECA 10 were 69.52/72.07, and among women with a self-reported history of genital warts, 60.09/66.73, respectively.

Conclusion There was no significant difference in the QoL of HPV vaccinated women between the two CC screening arms. Women were mostly satisfied with the screening experience despite the screening frequency. This information is important for the future screening program planning as we need to reach the best possible balance with screening benefits and harms. **Trial registration number** NCT02149030, date of registration 29/5/2014.

Keywords Human papillomavirus · HPV · HPV vaccination · Quality of life · Cervical cancer screening

Karolina Louvanto karolina.louvanto@tuni.fi

- ¹ Faculty of Medicine and Health Technology, Tampere University, Arvo Ylpön Katu 34, 33014 Tampere, Finland
- ² Department of Obstetrics and Gynecology, Tampere University Hospital, Tampere, Finland
- ³ FICAN-Mid, Tampere, Finland
- ⁴ Faculty of Social Sciences, Tampere University, Tampere, Finland
- ⁵ Department of Laboratory Medicine, Karolinska Institute, Stockholm, Sweden

Plain English summary

Vaccinations against human papillomavirus have been expanded to many countries during recent years, and the aim is to reduce the number of cervical cancers together with cervical cancer screening programs. The quality of life of vaccinated women may be affected by cervical cancer screening as screening and related examinations are known to cause anxiety and fear of cancer. Vaccinated women are coming to the screening age and there is a great need to design new screening programs for them. This study investigated the quality of life of vaccinated women in two different screening arms, frequently and infrequently screened. We found no difference in quality of life between the two screening arms. Cervical cancer screening seems to be well accepted among women and more frequent screening did not reduce the quality of life. On the other hand, vaccinated women seem to feel safe also despite the more infrequent screening as the quality of life was good also among women in this arm. In future, it will most likely be appropriate and safe to reduce the frequency of cervical cancer screening among vaccinated women.

Background

Human papillomaviruses (HPVs) are common DNA viruses that infect epithelial cells and cause a variety of epithelial lesions ranging from common warts to intraepithelial neoplasia and cancer [1]. IIt is widely acknowledged that approximately 80% of women will be infected with HPV during their lifetime and persistent high-risk (HR) HPV types are necessary for the development of all cervical cancer (CC) [2]. For HPV prevention, HPV vaccination has been introduced so far in more than 100 countries worldwide [3]. HPV vaccination has shown to be effective and safe [4]. In long term, together with CC screening programs, HPV vaccinations will greatly reduce HPV-related disease burden around the world [5, 6].

It is known that HPV-related diseases negatively affect the quality of life (QoL) of women in several ways [7, 8]. Diagnosed HPV infection can in worst case disrupt women's interpersonal and sexual relationships [9]. There is also increasing amount of knowledge on psychological adverse effects of cervical lesions and associated examinations and treatments. Abnormal Pap cytology, positive HPV test result, and colposcopy with or without biopsies or treatment are known to provoke significant anxiety on women. [10–13] Most worries concern the possibility of being diagnosed with CC, but also concerns on future fertility and sexuality have been commonly reported [14, 15]. Also, short-term decrease in sexual function has been reported after loop electrosurgical excision procedure (LEEP) of cervical dysplasia [16]. Younger women present higher concerns about their future, health, and fertility [15]. This might be due to the association of cervical HPV infection, precancer, and their treatments with increased risk of premature birth [17], and in some studies also with miscarriages [18].

The increasing interest in the personal viewpoint of patients' health has led to a demand for reliable and valid standardized questionnaires of health-related quality of life (HRQoL), and a variety of questionnaires have been developed to meet this need [19]. HRQoL is referring to patient's perceptions of the impact of disease and treatment on their physical, psychological, and social function and well-being [20].

In countries with established or emerging HPV vaccination programs and pre-existing screening, there is a great need to design the most optimal future cervical cancer screening program for HPV vaccinated women. This is currently highly crucial as the first birth cohorts of HPV vaccinated women are approaching the screening age in many countries. Among HPV vaccinated women, severe HPV-related diseases will occur more rarely compared to unvaccinated women, and a highly reduced screening program, even only once in a lifetime screening, may be enough and safe [21]. With the current study, our aim was to compare the HRQoL between frequently and infrequently screened HPV vaccinated women and provide information on CC screening related QoL to support future screening planning.

Methods

Participants

The current study consists of 753 HPV16/18 vaccinated women who originally participated in the communitybased Finnish HPV vaccination trial as previously described [22]. Bivalent HPV vaccine was given to the participants at the age of 13–15. At the age of 22, they were re-randomized to frequent (FS) and infrequent screening (IFS) arms. All women in both arms came to the screening visits three times during the study period, at the age of 22, 25, and 28. Cervical samples were taken from all women during every visit. Women in frequent screening arm got to know their screening results after every screening visit, while women in infrequent screening arm got to know their results only after their last follow-up visit. As an exception, also infrequently screened were informed about their earlier results if they were indicative of colposcopy referral (ASC-H, HSIL, or AGC in Pap cytology). At the age of 28, a survey considering background information, opinions on screening and HRQoL were sent to participants in both screening arms after their latest CC screening results. The validated HRQoL questionnaires included EuroQol Visual Analogue Scale (EQ VAS), the Finnish version of the RAND 36-Item Health Survey (RAND 36), and amended CECA (a Spanish acronym for the Specific Questionnaire for Condylomata Acuminata, Cuestionario Específico para Condiloma Acuminado) that are described in detailed under.

EuroQol Visual Analogue Scale (EQ VAS)

EQ VAS is a standardized, validated generic questionnaire in which the patient describes the experienced general state of health in a scale from 0 indicating worst imaginable health state to 100 indicating best imaginable health state [23].

Finnish version of the RAND 36-Item Health Survey (RAND 36)

RAND 36 is a standardized, validated generic questionnaire which assesses health and well-being in eight different domains: Role limitations due to physical health problems, Physical functioning, Pain, General health, Energy, Social functioning, Role limitation due to emotional problems, and Emotional well-being. RAND 36 scores range from 0 indicating worst HRQoL to 100 indicating best HRQoL. Scoring is based on answers to combinations of questions. [24]

Cuestionario Específico para Condiloma Acuminado, Spanish acronym for the Specific Questionnaire for Condylomata Acuminata (CECA 10)

CECA 10 is a disease-specific HRQoL questionnaire for patients with genital warts [25]. Earlier studies have shown that the CECA 10 questionnaire is a valid, reliable, and sensitive tool for the assessment of HRQoL. The questions refer to the past 7 days. Questionnaire comprises 10 items and 2 dimensions. The emotional dimension includes 6 items, and the sexual activity dimension includes 4 items. Each question allows an answer on a 5-option Likert scale indicating the degree of agreement: 1 = "totally agree," 2 = "agree," 3 = "don't know," 4 = "disagree," and 5 = "totally disagree." The global scoring range is 10–50, ranging from 6 to 30 in the emotional dimensions and from 4 to 20 in the sexual activity dimension. All dimensions are standardized for a scoring between 0 indicating worst HRQoL and 100 indicating best HRQoL. [26] In our study CECA 10 was minimally modified and therefore used for women with a self-reported history of Pap cytology abnormalities in addition to women with a self-reported history of genital warts.

Statistical analysis

RAND 36 questionnaire was modified in our study as Physical functioning -domain was omitted from the questionnaire because it wouldn't have been usable among the young population of this study. RAND 36 scores were calculated following the general scoring instructions of the questionnaire [24]. Scores were calculated for each domain if a woman had answered to at least half of the questions of the current domain and had answered to all questions of the domain of Pain and Social functioning. Otherwise, women were excluded from the RAND 36 analysis. In our study, amended CECA 10 questionnaire was intended only for women who reported a history of Pap cytology abnormalities or genital warts in the survey. Screening results were not verified in this context by the researchers. Emotional and sexual activity dimension as well as overall scores were calculated separately for each dimension using those that had a reported answer. Scores for each dimension were calculated only for women who had answered all questions of the dimension.

Survey included five questions concerning women's opinion about CC screening, more precisely on whether Pap cytology sampling was a pleasant experience, whether women were sufficiently informed before Pap cytology sampling and whether they knew how to get the test results. Questions are shown in Table 2 with results. Each question allowed an answer on a 5-option Likert scale: 1 = "absolutely true," 2 = "mostly true," 3 = "don't know," 4 = "mostly not true," and 5 = "absolutely not true." In questions 1-4, answers 1 and 2 (1 = "absolutely true" and 2 = "mostly true") were combined to illustrate satisfaction on Pap cytology sampling. In question 5, answers 4 and 5 (4 = "mostly not true," 5 = "absolutely not true") were combined to illustrate satisfaction.

Statistical analyses were performed using Stata version 17.0. FS and IFS arms were compared using statistical tests suitable for each variable. P value ≤ 0.05 was considered statistically significant. Chi-squared or Mann–Whitney rank-sum test, and if the expected frequency in each cell was below five in at least 80% of the cells, Fischer exact test, was used to compare the background characteristics and different scores among the women in FS and IFS arms.

Results

A total of 753 responses out of 1808 invitations (41.6%) were received, of which 370 were from the FS arm and 383 from the IFS arm. Majority of the women in both arms were from the birth cohort of 1993. Sociodemographic, as well as clinical characteristics and different sexual behaviors of the women are shown in Table 1. There was no significant difference between the two arms in most of the background characteristics. Small differences were observed in the proportions of different birth control methods used between the women in different study arms, as interrupted intercourse was more common method of pregnancy prevention among frequently screened than infrequently screened women (37.4% vs. 30.0%, p=0.050). The opposite was observed for hormonal IUD usage (21.8% vs. 28.1%, p = 0.035). Also, the proportion of women who had had oral sex with the latest partner was higher among the infrequently screened women (82.8% vs. 89.2%, p=0.013) but no difference was found in the frequency of oral sex protection use habits between the arms. Women in the frequently screened arm reported more often having a history of Pap cytology abnormalities (21.2%) vs. 14.0%, p = 0.010). On the other hand, no difference in the reported history of genital warts was observed.

EQ VAS scores that evaluated the experienced general state of health did not show a significant difference between

Table 1Backgroundinformation of the frequentlyand infrequently screened HPVvaccinated women at the ageof 28. Statistically significantresults are shown in bold^a

	Frequently screened n=370		$\frac{\text{Infrequently}}{n=383}$		p value
	n	%	n	%	
Year of birth					0.853
1992	66	17.8	67	17.5	
1993	231	62.4	246	64.2	
1994	73	19.7	70	18.3	
Working/studying					0.881
Working	248	69.5	261	70.2	
Studying	36	10.1	32	8.6	
Not working or studying	32	9.0	32	8.6	
Working and studying	41	11.5	47	12.6	
Study place		11.5	.,	12.0	0.523
Higher education	43	53.8	44	53.7	0.020
Polytechnic	22	27.5	27	32.9	
Upper secondary vocational school	10	12.5	5	6.1	
Apprenticeship	10	1.3	3	3.7	
Other	4	5.0	3	3.7	
Contraceptive method	+	5.0	5	5.7	
None	0	0	2	0.5	0.499
Combined contraceptive product $(E+P)$ or progestin pills	337	94.1	351	93.1	0.783
Progestin implant	32	8.9	29	93.1 7.7	0.785
Condom	351	98.0	359	95.2	0.588
					0.304
Pessary, spermicides	1 134	0.3 37.4	5	1.3 30.0	0.217
Interrupted intercourse			113	50.0 62.3	
Emergency contraception Hormone IUD	213 78	59.5	235	02.5 28.1	0.290
		21.8	106		0.035
Copper IUD	17	4.7	17	4.5	0.918
Smoking	225	())	224	(1.4	0.495
Never smoked	235	64.4	234	61.4	
Former smoker	72	19.7	74	19.4	
Current smoker	58	15.9	73	19.2	0.220
Married or in a permanent relationship	285	77.2	307	80.2	0.328
Have had sexual intercourse	358	96.8	376	98.4	0.134
Age at first sexual intercourse (years)		10.6	50	15.0	0.487
12–14	66	18.6	59	15.8	
15–17	177	49.9	203	54.3	
18–20	85	23.9	93	24.9	
21–23	21	5.9	15	4.0	
≥24	6	1.7	4	1.1	
Number of lifetime sexual partners	10				0.717
1	40	11.3	44	11.7	
2	26	7.3	33	8.8	
3	23	6.5	22	5.9	
4	16	4.5	26	6.9	
5–9	97	27.3	96	25.6	
≥ 10	153	43.1	154	41.1	0.348
Number of sexual acts during past year with the most recent sexual partner					
0	14	4.1	15	4.2	
1–20	113	32.9	116	32.1	

Table 1 (continued)

	Frequently screened n=370		$\frac{\text{Infrequently}}{n=383}$		p value
	\overline{n}	%	n	%	
21–50	109	31.8	113	31.3	
51–100	73	21.3	94	26.0	
>100	34	9.9	23	6.4	
Number of sexual acts during past year with previous sexual partner					
0	66	36.5	67	38.5	
1–20	77	42.5	65	37.4	
21–50	20	11.0	27	15.5	
51-100	13	7.2	13	7.5	
>100	5	2.8	2	1.1	
Frequency of condom use with the most recent partner					0.072
Never	177	51.2	204	56.2	
Half of the times	35	10.1	50	13.8	
Every time	78	22.5	67	18.5	
Other	56	16.2	42	11.6	
Practicing oral sex with the most recent partner	293	82.8	323	89.2	0.013
Using oral sex protection with the most recent partner	10	3.4	9	2.8	0.653
Self-reported history of Pap cytology abnormalities	76	21.2	53	14.0	0.010
Self-reported history of genital warts	31	9.0	33	9.1	0.943

E estrogen, P progestin

 a_p value ≤ 0.05

Table 2 RAND 36-Item Health Survey $^{\rm a}$ scores of frequently and infrequently screened HPV vaccinated women at the age of 28

	Frequently screened n=370		Infrequently screened n = 383		p value
	Mean	SD	Mean	SD	
Physical role functioning	78.13	32.23	81.64	30.47	0.128
Pain	77.93	20.78	80.18	19.48	0.128
General health	69.10	19.00	69.12	18.65	0.989
Energy	54.67	19.46	53.61	20.43	0.471
Social functioning	83.72	19.67	85.11	20.00	0.344
Emotional role functioning	69.53	38.47	69.68	36.47	0.956
Emotional well-being	68.16	18.17	69.29	17.49	0.385

SD standard deviation

p value ≤ 0.05 was considered statistically significant

^aPhysical functioning domain omitted

FS and IFS arm. In a scale from 0 to 100, median EQ VAS was 80 (Q_1 – Q_3 75–90) in both arms (data not shown).

RAND 36 results are shown in Table 2. There was no significant difference between the two arms in any domain. However, a trend of lower scores was seen in six of seven

domains among frequently screened women compared to infrequently screened. Energy domain was an exception to this trend.

Amended CECA 10 scores of women with a self-reported history of Pap cytology abnormalities or genital warts are shown in Table 3. A total of 106 (14.1%) women had answered to the amended CECA 10 questionnaire which was intended for women with a history of Pap cytology abnormalities. There was no significant difference in amended CECA 10 scores between the two arms. However, a trend of lower scores was seen in all amended CECA 10 dimensions among frequently screened women compared to infrequently screened. In addition, a total of 55 (7.3%) had answered to the amended CECA 10 questionnaire intended for women with a history of genital warts. There was no significant difference in amended CECA 10 scores between the arms. Again, a trend of lower scores among frequently screened women was observed in emotional and overall scores, though.

Five questions in the survey concerned the women's opinion about CC screening. Results are shown in Table 4. There was no significant difference between the two arms in the proportion of women who were satisfied with the Pap cytology sampling. Depending on the specific question, proportions of satisfied women were 73.4–97.5% and 74.3–96.8% in FS and IFS arms, respectively. The survey included one

 Table 3
 Amended
 CECA 10 scores of frequently and infrequently screened
 HPV vaccinated women who reported a history of Pap cytology abnormalities or genital warts

	Frequently screened		Infrequently screened		p value		
	Mean	SD	Mean	SD			
Pap cytology abnormalities (n (FS/IFS)=64/42)							
Emotional dimension	63.22	27.59	64.23	22.13	0.877		
Sexual dimension	80.00	25.39	84.23	17.74	0.822		
Overall	69.52	24.09	72.07	17.56	0.792		
Genital warts (n (FS/IFS) = $29/26$)							
Emotional dimension	55.32	30.62	65.87	24.92	0.190		
Sexual dimension	68.29	30.32	68.03	28.63	0.885		
Overall	60.09	28.16	66.73	25.01	0.359		

CECA 10 scores range from 0 (the worst HRQoL) to 100 (the best HRQoL)

CECA Spanish acronym of specific questionnaire for condylomata acuminate, FS frequently screened, IFS infrequently screened, SD standard deviation

p value ≤ 0.05 was considered statistically significant

question about overall satisfaction in the current life situation at the age of 28. In a scale from 0 to 10, median score in both arms was 8 (Q_1-Q_37-9) (data not shown.)

Discussion

These results are, to our knowledge, the first to evaluate the effect of screening frequency on the QoL of HPV vaccinated women. It has been expected that HPV vaccinations would improve HRQoL by reducing incidence of HPV-related lesions. However, earlier studies have shown that CC screening lowers the HRQoL in long term.

In our study, the frequency of CC screening did not significantly affect the HRQoL of young HPV vaccinated women. At first, this seems quite unexpected, as in previous studies among unvaccinated women, the history of genital warts or cytological abnormalities, the conditions commonly related to frequent smear sampling and healthcare visits have been associated with reduced QoL [7, 8, 27-29]. Especially anogenital warts, caused by low-risk HPV types, have shown to cause psychological distress by making patients feel more ashamed of themselves and less attractive. In addition, diagnosis and treatment of genital warts have shown to trigger high level of anxiety, anger, and depression, and approximately, two-thirds of patients have made lifestyle changes especially regarding their sexual relationships [9]. Moreover, women with anogenital warts are shown to have fears about sexual transmission which affects their self-image and life control perception [7, 30], and sexual enjoyment and activity are negatively affected [31]. However, no difference has been yet reported between HPV vaccinated and unvaccinated women [8, 32]. In our study, neither the rates of self-reported anogenital warts, nor the number of actual cervical screening samplings differed between study arms, which is one of the probable explanations behind the similar HRQoL scores.

Earlier studies have suggested that cervical lesions as well as associated examinations and treatments cause psychological adverse effects [10–12]. During the current study, participating women met a health care professional regularly and had therefore the opportunity to ask questions and receive advice. This may have lessened their concerns about HPVrelated diseases and made them feel safe and taken care. Thus, also frequently screened women may have stayed less anxious when waiting for and getting the screening results. However, a trend of lower scores was seen in amended CECA 10 and RAND 36 scores among frequently screened women compared to infrequently screened but the difference was not statistically significant. These results support the hypothesis that women might be more distressed of repeated screening. It is possible, that if the length of follow-up were longer and the difference between study arms in the number of screening visits were wider, also differences between HRQoL indicators would have been observed. Further, it is

 Table 4
 Proportions of frequently and infrequently screened HPV vaccinated women who were satisfied with the Pap cytology sampling experience

	Frequently screened	Infrequently screened	p value
	n=370	n=383	
	%	%	
Pap cytology sampling was considered as a pleasant experience	73.4	74.3	0.803
Enough information about sampling was given before the visit	94.0	93.2	0.658
Women knew how to prepare for sampling before the visit	93.2	93.7	0.770
Enough advice and guidance were received regarding the sampling situation	97.5	96.8	0.660
Women were aware of how and when she would get the Pap cytology test result	83.7	86.4	0.307

p value ≤ 0.05 was considered statistically significant

worth noticing that all the participating women had accepted as adolescents to participate HPV vaccination trial and might have more positive attitude toward health care interventions as general population, which may have diluted the difference in HRQoL between the arms.

Disease-specific questionnaires that only contain items specifically designed for a particular condition are more likely to be relevant and sensitive to patients in areas that clinicians may wish to monitor [25]. The CECA 10 questionnaire has been developed and validated for genital warts and not for cytological abnormalities. However, earlier studies have used it also in this patient group and it has been suggested relevant [8]. In our study, the survey was sent after the women had received their results from the latest screening. There was no significant difference in amended CECA 10 scores between the two arms among women with a self-reported history of genital warts or Pap cytology abnormalities. The history of these anogenital or cervical changes was based on the women's answers in the survey. Screening results were not verified in this context by the researchers. Earlier studies have shown that HPV-related QoL is worse at initial diagnosis than in later assessment [33]. This is understandable as the HPV infection commonly disappears within 1-2 years. This may influence our results since patients were already at 28 years of age and some earlier Pap cytology abnormalities and genital warts may have resolved before answering the survey.

Liddon et al. studied HPV vaccination uptake and sexual behavior among adolescent and young women. In their study, HPV vaccination was not associated with being sexually active or with number of sex partners. Among sexually active adolescents, vaccinated girls were more likely to always use a condom. [34] Studies on the effect of contraceptives on cervical dysplasia have been controversial [35, 36]. It is not known how this might influence women's contraceptive use. In our study, sexual activity was equal among vaccinated women in both screening arms considering the number of sexual partners and the number of sexual sessions during last year. In our vaccinated population, there was no difference in the frequency of condom use between the frequently and infrequently screened women. Interrupted intercourse was more common pregnancy control method in frequently screened arm. Women in the infrequently screened arm were more likely to use hormonal IUD. There was no significant difference in other hormonal contraceptive product use between the arms.

Our study has some limitations. The response rate was relatively low, which is typical for questionnaire studies, and selection bias may exist. It is possible that responded women had more motivation to take part to the study. However, the response rate was very similar in both arms, and therefore, we estimate that the possible selection bias is equal in both arms and did not impact much these results. The follow-up time was six years and CC screening samples were taken in three-year intervals. Short timeframes between the screening visits might have diluted the differences in the QoL between the study arms. More studies on how CC screening affects the QoL among HPV vaccinated women would be valuable to achieve more in-depth information on women's view on the screening experience. Longer follow-up time and including a reference arm of unvaccinated women with an additional depression scale questionnaire could give a more comprehensive understanding on the QoL and CC screening.

The current CC screening programs are in verge of changes as the first HPV vaccinated women are currently approaching the CC screening age in many countries. CC screening programs will need to be adjusted for HPV vaccinated women to be able to provide them the best possible screening tests and schedule. When planning future CC screening, it is important also to consider women's opinion about screening. In our study, there was no significant difference between the arms in the proportion of women who were satisfied with the Pap cytology sampling experience; both the frequently and infrequently screened women were mostly satisfied with screening. Thus, CC screening seems to be well accepted among HPV vaccinated women from the individual's point of view. HPV vaccinations reduce the prevalence of HR-HPV infections and cervical cancer, and vaccinated women may therefore feel safer also in infrequent screening program. This information is important for the future screening program planning for HPV vaccinated women. A highly reduced screening program is likely to be safe due to the lower prevalence of HPV-related diseases. We need to reach the best possible balance with screening benefits and harms. Cost-effectiveness analysis of screening must also be considered. More infrequent CC screening of HPV vaccinated women in future will most likely be costeffective and safe without a negative effect to the QoL.

Acknowledgements The preliminary results were presented in Eurogin International Multidisciplinary HPV Congress held in Bilbao, Spain, on February 10th, 2023.

Author Contributions ML, TE, and KL contributed to the study conception and design. Material preparation and data collection were performed by TE, AB, KH, KH, MH, and MN. Analyses were performed by KT, KL, and HH. The first draft of the manuscript was written by KT, and all authors contributed to the review, writing, and editing of the manuscript. All authors read and approved the final version of the manuscript.

Funding Open access funding provided by Tampere University (including Tampere University Hospital). This study was partly supported by Tampere University Hospital Support Foundation, Tampere University Hospital (Project No MK371) and The Finnish Medical Foundation (Grant No 5525).

Data availability Individual participant data that underlie the results reported in this article, after de-identification, will be available for

researchers who provide a methodologically sound proposal with achievable aims. Proposals should be directed via e-mail to the corresponding author (KL); data requestors will need to sign a data access agreement.

Declarations

Competing interests The authors have no relevant financial or non-financial interests to disclose.

Ethical approval This study was performed in line with the principles of the Declaration of Helsinki. Approval was granted by the Pirkanmaa Hospital District Ethical Review Board (Date 12/6/2007, 19/2/2014, 4/2/2020 / No R07113M, R13149, R191936).

Consent to participate Informed written consent was obtained from all individual participants included in the study.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

References

- Smith, J. S., Lindsay, L., Hoots, B., Keys, J., Franceschi, S., Winer, R., & Clifford, G. M. (2007). Human papillomavirus type distribution in invasive cervical cancer and high-grade cervical lesions: A meta-analysis update. *International Journal of Cancer*, 121(3), 621–632. https://doi.org/10.1002/ijc.22527
- Trottier, H., & Franco, E. L. (2006). The epidemiology of genital human papillomavirus infection. *Vaccine*, 24, S4–S15. https://doi. org/10.1016/j.vaccine.2005.09.054
- Markowitz, L. E., & Schiller, J. T. (2021). Human papillomavirus vaccines. *The Journal of Infectious Diseases.*, 224(Supplement_4), S367–S378. https://doi.org/10.1093/infdis/jiaa621
- Arbyn, M., & Xu, L. (2018). Efficacy and safety of prophylactic HPV vaccines. A cochrane review of randomized trials. *Expert Review of Vaccines*, 17(12), 1085–1091. https://doi.org/10.1080/ 14760584.2018.1548282
- Lehtinen, M., Paavonen, J., Wheeler, C. M., Jaisamrarn, U., Garland, S. M., Castellsagué, X., & Dubin, G. (2012). Overall efficacy of HPV-16/18 AS04-adjuvanted vaccine against grade 3 or greater cervical intraepithelial neoplasia: 4-year end-of-study analysis of the randomised, double-blind PATRICIA trial. *The Lancet Oncology*, *13*(1), 89–99. https://doi.org/10.1016/S1470-2045(11)70286-8
- Sung, H., Ferlay, J., Siegel, R. L., Laversanne, M., Soerjomataram, I., Jemal, A., & Bray, F. (2021). Global cancer statistics 2020: GLOBOCAN estimates of incidence and mortality worldwide for 36 cancers in 185 countries. *CA: A Cancer Journal for Clinicians*. https://doi.org/10.3322/caac.21660
- Wang, S.-M., Shi, J.-F., Kang, D.-J., Song, P., & Qiao, Y.-L. (2011). Impact of human papillomavirus-related lesions on quality of life: A multicenter hospital-based study of women in Mainland

China. International Journal of Gynecologic Cancer, 21(1), 182– 188. https://doi.org/10.1097/IGC.0b013e3181ffbed8

- Woodhall, S., Eriksson, T., Nykänen, A.-M., Huhtala, H., Rissanen, P., Apter, D., & Lehtinen, M. (2011). Impact of HPV vaccination on young women's quality of life a five year follow-up study. *The European Journal of Contraception & Reproductive Health Care*, *16*(1), 3–8. https://doi.org/10.3109/13625187.2010. 536921
- Clarke, P., Ebel, C., Catotti, D. N., & Stewart, S. (1996). The psychosocial impact of human papillomavirus infection: Implications for health care providers. *International journal of STD & AIDS*, 7(3), 197–200. https://doi.org/10.1258/0956462961917618
- Jentschke, M., Lehmann, R., Drews, N., Hansel, A., Schmitz, M., & Hillemanns, P. (2020). Psychological distress in cervical cancer screening: results from a German online survey. *Archives of Gynecology and Obstetrics*. https://doi.org/10.1007/ s00404-020-05661-9
- McBride, E., Marlow, L. A. V., Forster, A. S., Ridout, D., Kitchener, H., Patnick, J., & Waller, J. (2020). Anxiety and distress following receipt of results from routine HPV primary testing in cervical screening: The psychological impact of primary screening (PIPS) study. *International Journal of Cancer*. https://doi. org/10.1002/ijc.32540
- Sharp, L., Cotton, S., Gray, N., Avis, M., Russell, I., Walker, L., & Little, J. (2011). Long-term psychosocial impact of alternative management policies in women with low-grade abnormal cervical cytology referred for colposcopy: A randomised controlled trial. *British Journal of Cancer*. https://doi.org/10.1038/ sj.bjc.6606042
- Gray, N. M., Sharp, L., Cotton, S. C., Masson, L. F., Little, J., Walker, L. G., & Woolley, C. M. (2006). Psychological effects of a low-grade abnormal cervical smear test result: Anxiety and associated factors. *British Journal of Cancer*. https://doi.org/10. 1038/sj.bjc.6603086
- Ideström, M., Milsom, I., & Andersson-Ellström, A. (2003). Women's experience of coping with a positive Pap smear: A register-based study of women with two consecutive Pap smears reported as CIN 1. Acta Obstetricia et Gynecologica Scandinavica, 82(8), 756–761. https://doi.org/10.1080/j.1600-0412. 2003.00165.x
- Sharp, L., Cotton, S. C., Cruickshank, M. E., Gray, N. M., Neal, K., Rothnie, K., & Wells, M. (2015). Long-term worries after colposcopy: Which women are at increased risk? *Women's Health Issues*. https://doi.org/10.1016/j.whi.2015.04.002
- Inna, N., Phianmongkhol, Y., & Charoenkwan, K. (2010). Sexual function after loop electrosurgical excision procedure for cervical dysplasia. *Journal of Sexual Medicine*. https://doi.org/ 10.1111/j.1743-6109.2009.01633.x
- Kyrgiou, M., Athanasiou, A., Paraskevaidi, M., Mitra, A., Kalliala, I., Martin-Hirsch, P., & Paraskevaidis, E. (2016). Adverse obstetric outcomes after local treatment for cervical preinvasive and early invasive disease according to cone depth: Systematic review and meta-analysis. *BMJ (Online)*. https://doi.org/10. 1136/bmj.i3633
- Popescu, S. D., Boiangiu, A. G., Sima, R.-M., Bilteanu, L., Vladareanu, S., & Vladareanu, R. (2022). Maternal HPV infection and the estimated risks for adverse pregnancy outcomes—a systematic review. *Diagnostics*, 12(6), 1471. https://doi.org/10. 3390/diagnostics12061471
- Costa, D. S. J., Mercieca-Bebber, R., Rutherford, C., Tait, M.-A., & King, M. T. (2021). How is quality of life defined and assessed in published research? *Quality of Life Research*, 30(8), 2109–2121. https://doi.org/10.1007/s11136-021-02826-0
- Guyatt, G. H., Feeny, D. H., & Patrick, D. L. (1993). Measuring health-related quality of life. *Annals of internal medicine*,

118(8), 622–629. https://doi.org/10.7326/0003-4819-118-8-199304150-00009

- Louvanto, K., Eriksson, T., Gray, P., Apter, D., Baussano, I., Bly, A., & Lehtinen, M. (2020). Baseline findings and safety of infrequent vs frequent screening of human papillomavirus vaccinated women. *International Journal of Cancer*, 147(2), 440–447. https://doi.org/10.1002/ijc.32802
- Lehtinen, M., Apter, D., Eriksson, T., Harjula, K., Hokkanen, M., Natunen, K., & Struyf, F. (2021). Effectiveness of various human papillomavirus vaccination strategies: A community randomized trial in Finland. *Cancer Medicine*, 10(21), 7759–7771. https://doi.org/10.1002/cam4.4299
- The EuroQol Group. (1990). EuroQol a new facility for the measurement of health-related quality of life. *Health Policy*, 16(3), 199–208. https://doi.org/10.1016/0168-8510(90)90421-9
- Aalto, A.-M., Aro, A. R., & Teperi, J. (1999). Rand-36 terveyteen liittyvän elämänlaadun mittarina : Mittarin luotettavuus ja suomalaiset väestöarvot.
- Badia, X., Colombo, J. A., Lara, N., Llorens, M. A., Olmos, L., Los de Terreros, M. S., & Vilata, J. J. (2005). Combination of qualitative and quantitative methods for developing a new health related quality of life measure for patients with anogenital warts. *Health and Quality of Life Outcomes*, 3(1), 24. https://doi.org/10. 1186/1477-7525-3-24
- Vilata, J. J., Varela, J. A., Olmos, L., Colombo, J. A., Llorens, M. A., De Los, S., Terreros, M., & Badía, X. (2008). Validation and clinical use of the CECA, a disease-specific quality of life questionnaire for patients with anogenital condylomata acuminata. *Acta dermato-venereologica*, 88(3), 257–262. https://doi.org/10. 2340/00015555-0422
- Paradisi, A., Capizzi, R., Ricci, F., di Pietro, C., Abeni, D., & Tabolli, S. (2013). Quality of life in patients with anogenital warts. *European Journal of Dermatology*, 23(6), 837–842. https://doi. org/10.1684/ejd.2013.2171
- Maggino, T., Casadei, D., Panontin, E., Fadda, E., Zampieri, M. C., Donà, M. A., & Altoè, G. (2007). Impact of an HPV diagnosis on the quality of life in young women. *Gynecologic Oncology*, *107*(1), S175–S179. https://doi.org/10.1016/j.ygyno.2007.07.013
- Pereira-Caldeira, N. M. V., Góes, F. G. B., Almeida-Cruz, M. C. M., Caliari de, J. S., Pereira-Ávila de, F. M. V., & Gir, E. (2020). Quality of life for women with human papillomavirus-induced lesions. *Revista Brasileira de Ginecologia e Obstetrícia / RBGO*

Gynecology and Obstetrics, 42(04), 211–217. https://doi.org/10. 1055/s-0040-1709192

- Nagele, E., Reich, O., Greimel, E., Dorfer, M., Haas, J., & Trutnovsky, G. (2016). Sexual activity, psychosexual distress, and fear of progression in women with human papillomavirus-related premalignant genital lesions. *The Journal of Sexual Medicine*, *13*(2), 253–259. https://doi.org/10.1016/j.jsxm.2015.12.012
- Ireland, J. A., Reid, M., Powell, R., & Petrie, K. J. (2005). The role of illness perceptions: Psychological distress and treatmentseeking delay in patients with genital warts. *International Journal* of STD & AIDS, 16(10), 667–670. https://doi.org/10.1258/09564 6205774357334
- Eriksson, T., Torvinen, S., Woodhall, S. C., Lehtinen, M., Apter, D., Harjula, K., & Lehtinen, M. (2013). Impact of HPV16/18 vaccination on quality of life: A pilot study. *The European Journal* of Contraception & Reproductive Health Care, 18(5), 364–371. https://doi.org/10.3109/13625187.2013.801953
- Hsu, Y.-Y., Wang, W.-M., Fetzer, S. J., Cheng, Y.-M., & Hsu, K.-F. (2018). Longitudinal psychosocial adjustment of women to human papillomavirus infection. *Journal of Advanced Nursing*, 74(11), 2523–2532. https://doi.org/10.1111/jan.13725
- Liddon, N. C., Leichliter, J. S., & Markowitz, L. E. (2012). Human Papillomavirus Vaccine and Sexual Behavior Among Adolescent and Young Women. *American Journal of Preventive Medicine*, 42(1), 44–52. https://doi.org/10.1016/j.amepre.2011.09.024
- Anastasiou, E., McCarthy, K. J., Gollub, E. L., Ralph, L., van de Wijgert, J. H. H. M., & Jones, H. E. (2022). The relationship between hormonal contraception and cervical dysplasia/cancer controlling for human papillomavirus infection: A systematic review. *Contraception*, 107, 1–9. https://doi.org/10.1016/j.contr aception.2021.10.018
- Asthana, S., Busa, V., & Labani, S. (2020). Oral contraceptives use and risk of cervical cancer—a systematic review & metaanalysis. *European Journal of Obstetrics & Gynecology and Reproductive Biology*, 247, 163–175. https://doi.org/10.1016/j. ejogrb.2020.02.014

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.